

Comment:

I suggest giving some examples in section c) Early Detection of Emerging Diseases and d) Rare or Unusual Cases to give readers a deeper impression.

Reply:

Dear Reviewer,

Thank you for your valuable feedback and suggestions

I have included the examples for the above 2 sections, as you suggested for a better & deeper impression.